Author's response to reviews

Title: Multifocal invasive ductal breast cancer with osteoclast-like giant cells

Authors:

Georg Richter (richter@pathologie-hannover.de)
Christoph Uleer (christoph.uleer@t-online.de)
Thomas Noesselt (noesselt@kreiskrankenhaus-hameln.de)

Version: 2 Date: 31 May 2010

Author's response to reviews:

Dear Ladies and Gentlemen,

following Your instructions I restructure the abstract and include the ethnicity. In addition, I include a conclusion and removed strange subheadings. You can find our confirmation, a competing interests section, an authors contribution section and the figure titles.

Yours faithfully

Georg Richter, MD